Author’s response to reviews

Title: Predictors of Provider-Initiated HIV Testing and Counseling refusal by outpatient department clients in Wolaita zone, Southern Ethiopia: A case control study

Authors:

Wolde Faltamo (woldiefacha@gmail.com)

Wondwosen Kassahun (wondkyt@yahoo.com)

Abdulhalik Workicho (abdulhalikw@gmail.com)

Version: 2 Date: 30 May 2016

Author’s response to reviews:

Dear Editor,

I have received your email once again. I would like to thank you in advance for considering our manuscript for publication on BMC public health. I have seen the comments indicated by both of you (the reviewer and the editor). Hence, I addressed the comments in my revised manuscript as well and you can differentiate easily as I colored it.

Response to Reviewer reports:

1. Professional Copyediting

We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Response to #1. The language part was exhaustively edited and you can see it from the manuscript.
2. Please provide a 'Declarations' section in your manuscript as detailed here: http://bmcpublichealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

After the main sections in the body of the manuscript, all article types must contain the following items under 'Declarations':

* Ethics and Consent to Participate statement
* Consent to Publish statements
* Consent forms should not be attached
* Competing Interest statement
* Au Contributions (if there is more than one author)
* Availability of data and materials statement
* Abbreviations (if present)

-Ethics (and Consent to Participate)

Please include a statement on ethics approval and consent to participate. All manuscripts describing studies involving human participants, human data, or human tissue require need to include a statement on ethics approval and consent (even where the need for approval was waived). Please provide the name of the ethics committee that approved the study and the committees reference number if appropriate. Studies involving animals must also include a statement on ethics approval.

If your manuscript does not report on or involve the use of any animal or human data or tissue, this section, please state Not applicable in this section.

Manuscripts which do not adhere to our editorial policies may experience delays or be rejected. Further information about our editorial policies can be found at the following link: http://www.biomedcentral.com/about/editorialpolicies#Ethics
- Consent to Publish

Please include a Consent for publication section in your manuscript. If your manuscript contains any individual persons data in any form (including individual details, images or videos), consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian. All presentations of case reports must have consent to publish. You can use your institutional consent form or our consent form if you prefer. You should not send the form to us on submission, but we may request to see a copy at any stage (including after publication).

If your manuscript does not contain any individual persons data, please state Not applicable in this section.

Consent form template: http://old.biomedcentral.com/download/consent_form.pdf

Ethical approval and consent: http://www.biomedcentral.com/about/editorialpolicies#Ethics

-Availability of Data and Materials

Please include an Availability of data and materials statement section in your article detailing where the data supporting your findings can be found. BioMed Central strongly encourages all datasets on which the conclusions of the manuscript rely to be either deposited in publicly available repositories (where available and appropriate) or presented in the main paper or additional supporting files, in machine-readable format (such as spreadsheets rather than PDFs) whenever possible.

However, if you do not wish to share your data, please state that data will not be shared, and state the reason.

Availability of data and materials: http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials
Response to #2. Declaration section was corrected as indicated by reviewers.

3. Please include a 'Conclusions' section in your manuscript.

Response to #3 Conclusion section was included in the manuscript.